Author’s response to reviews

Title: Even the best laid plans: a qualitative study of older adults’ experience transitioning to wheelchair use

Authors:

Edward M Giesbrecht (Ed.Giesbrecht@med.umanitoba.ca)
William C Miller (bill.miller@ubc.ca)
Roberta L Woodgate (Roberta.Woodgate@umanitoba.ca)

Version: 2 Date: 19 August 2014

Author’s response to reviews: see over
August 19, 2014

Danica Stephanie Lorrein Jose
Journal Editorial Office
BioMed Central

MS: 6966449471373914
Research article: Even the best laid plans: a qualitative study of older adults’ experience transitioning to wheelchair use

Dear Danica Jose:

Thank you for your response to our manuscript submission and request for revisions prior to peer review. The following is a summary of the revisions made, as requested:

1. *I would kindly ask the authors to formulate a research question.*
   The background section was revised to more explicitly present the research question (lines 96-101).

2. *Describe in more details data analysis and the process of selecting themes for the result presentation.*
   The data analysis section is now partitioned (with a separate heading) and additional detail introduced regarding process/procedures and theme selection (lines 146-161).

I trust that this meets with your approval. Please do not hesitate to contact me if you have further questions or concerns.

Kind regards;

Ed Giesbrecht